Author’s response to reviews

Title: INCREASING RETENTION OF HIV POSITIVE PREGNANT AND BREASTFEEDING MOTHERS ON OPTION-B PLUS BY UPGRADING AND PROVIDING FULL TIME HIV SERVICES AT A LOWER HEALTH FACILITY IN RURAL UGANDA

Authors:

Enos Masereka (mirembeenos@gmail.com)
Tom Ngabirano (tomngabirano@gmail.com)
Charles Osingada (chaposingada@gmail.com)
Christine Sekaggya (csekaggya@idi.co.ug)
Barbara Castelnuovo (bcastelnuovo@idi.co.ug)
Agnes Kiragga (akiragga@idi.co.ug)

Version: 7 Date: 29 Jun 2019

Author’s response to reviews:

1. We note that your manuscript contains potentially identifiable information; i.e. age, sex, etc., after each included participant quote. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, we ask that you update your Consent for publication statement to reflect whether explicit consent for publication of any and all potentially identifiable information has been obtained in writing from all participants. If consent for publication has not been obtained in writing then we ask that you please amend the manuscript to address this issue by removing the sex and age of each participant. Alternatively, you may opt to change them to randomized participant identifier numbers at this time in order to fully protect and ensure the anonymity of the participants.

*The manuscript has been improved by removing participant identifiable information*